CST Part IA Regular Languages and Finite Automata List of corrections to the lecture notes June 5, 2000

Corrections are underlined or crossed out, if not otherwise commented upon. Page 4, line -1: Read ŞorŤ for ŞnorŤ.

Page 12, line -7: Ş . . . to which one puts ^nite automataŤ.

Page 20, line 11: Şfrom the fact that q_1 ∈ S_1 = δPM (sPM , a_1) we deduce . . . Ť. Page 20, Exercise 2.5.1: ŞFor each of the two languages . . . Ť.

Page 33, lines -11,-10: Şby part (a) of Kleene’s Theorem . . . . Then by part (b) of the theorem . . . Ť. Page 37, Slide 28: Ş{(, ), a, b, . . . , z}Ť.

Page 43, Slide 34: Read Ş(where n ≥ 0)Ť for Ş(where n ≥ 1)Ť. Page 43, Exercise 5.4.2: Şconsider the DFA M^0 with . . . Ť.